
\documentclass[11pt,reqno]{amsart}
\usepackage[cp1251]{inputenc}
\usepackage[english,russian]{babel}
\usepackage{amsmath}
\usepackage{amssymb}
\usepackage{amsfonts}
\usepackage{graphicx}
\usepackage{xcolor}
\usepackage[colorlinks]{hyperref}



\usepackage{titlesec}
\titlespacing{\section}{0pt}{2.5ex}{1.5ex}
\titlespacing{\subsection}{0pt}{1.5ex}{1ex}
\titlespacing{\subsubsection}{0pt}{1.5ex}{1ex}
\titleformat{\section}{\large\bfseries\centering}{\thesection}{1em}{}
\titleformat{\subsection}[runin]{\bfseries}{\thesubsection.}{0.5em}{}[.\mbox{\ }]
\titleformat{\subsubsection}[runin]{\bfseries}{\thesubsubsection.}{0.4em}{}[.\mbox{\ }]





%%%%%---------- ORCiD
\newcommand\orcidicon[1]{\href{https://orcid.org/#1}{\includegraphics[scale=0.02]{orcid.pdf}}}


\def\udcs{510.67} %Here the author places classificators of the paper according to Russian classification system
\def\mscs{03C15, 03C64} %Here the author places  classificators according to the AMS classification list.
\setcounter{page}{144}

\newtheorem{lemma}{Lemma}
\newtheorem{theorem}{Theorem}
\newtheorem{definition}{Definition}
\newtheorem{corollary}{Corollary}
\newtheorem{proposition}{Proposition}



\hypersetup{linkcolor=blue,filecolor=black,urlcolor=blue, citecolor=blue}

\def\logo{\textcolor{blue}{{\bf\Huge S\raisebox{0.2ex}{\hspace{0.55ex}\raisebox{0.05ex}e\hspace{-1.65ex}$\bigcirc$}MR}}}

\def\semrtop
{
	\vbox{
		\noindent\logo \hspace{-3mm}  \parbox{12cm}{     \begin{center}
				{\LARGE \textsf{СИБИРСКИЕ ЭЛЕКТРОННЫЕ}} \\[2mm]
				{\LARGE \textsf{МАТЕМАТИЧЕСКИЕ ИЗВЕСТИЯ}} \\[2mm]
				{\Large \textsf{Siberian Electronic Mathematical Reports}} \\[1mm]
				{\Large\tt{\textcolor{blue}{http://semr.math.nsc.ru}}}\\[1mm]
				\raisebox{1ex}{ISSN 1813-3304 }
				\vspace*{-3mm}
				%     {\small Институт математики им. С.Л.Соболева СО РАН}\\[-1mm]
				%     {\small Sobolev Institute of Mathematics SB RAS}
			\end{center}
		}
		\vspace{-1mm}
		\noindent
		\begin{tabular}{c}
			\hphantom{aaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaaa!} \\
			\hline\hline
		\end{tabular}
		
		\vspace{3mm}
		{\flushleft\it Vol. 20, No. 2, pp. 144--145 (2023) \hspace{\fill}{\rm\small УДК \udcs \hspace{-0.3mm}}}
		\newline  %Volume, pages in Russian, filled by Editorial board
		{\rm\small \textcolor{blue}{https://doi.org/10.33048/semi.2023.20.???}}\hphantom{aaaaaaaaaaaaaaaa\;}{\rm\small MSC\ \mscs }
	}%\vbox
}



\begin{document}
\begin{thebibliography}{1}

\bibitem{Dries} L. van den Dries, \href{https://doi.org/10.1016/S0049-237X(08)71811-1}{\it Remarks on Tarski's problem concerning ($\mathbb{R}, +, *$, exp)}, Proc. Logic Colloquim' 82, (1984), 240--266.
\bibitem{MS} D. Marker, Ch. Steinhorn, \href{https://doi.org/10.2307/2275260}{\it Definable types in o-minimal Theories}, The Journal of Symbolic Logic, \textbf{59}:1 (1994), 185--198.
\bibitem{MMS} D. Macpherson, D. Marker, and Ch. Steinhorn, \href{https://doi.org/10.1090/S0002-9947-00-02633-7}{\it Weakly o-minimal structures and real closed fields}, Trans. Amer. Math. Soc., \textbf{352}:12 (1993), 5435--5483.
\bibitem{BCons} B.S. Baizhanov, \href{https://www.mathnet.ru/rus/vngu/v7/i3/p13}{\it Conservative extensions of models of weakly o-minimal theories}, Bulletin of Novosibirsk State University, \textbf{7}:3 (2007), 13--44. (in Russian) 
\bibitem{BDef} B.S. Baizhanov, \href{https://www.mathnet.ru/rus/mt/v8/i2/p3}{\it Definability of 1-types in weakly o-minimal theories}, Mathematical Works, \textbf{8}:2 (2005), 3--38. (in Russian)
\bibitem{BOtypes} B.S. Baizhanov, \href{https://scholar.google.com/scholar_lookup?title=One-types%20in%20weakly%20o-minimal%20theories&pages=75-88&publication_year=1996&author=Baizhanov%2CB%20S}{\it One-types in weakly o-minimal theories}, Proceedings of Informatics and Control Problems Institute, Almaty (1996), 75--88. %%https://goo.by/k1QxF
\bibitem{Shelah} S. Shelah \href{https://doi.org/10.1016/0022-4049(91)90078-G}{\it Classification theory and the number of non-isomorphic models, Second Edition}, Studies in Logic and the Foundations of Mathematics 92, Amsterdam: North-Holland (1990).
\bibitem{BClass} B.S. Baizhanov {\it Classification of one-types in weakly o-minimal theories and its corollaries}, Preprint. Institute of Informatics and Control Problems of Kazakh Academy of Sciences, Almaty, 1996.
\bibitem{Kul1} B.Sh. Kulpeshov \href{http://www.ams.org/mathscinet-getitem?mr=2028157}{\it The convexity rank and orthogonality in weakly o-minimal theories}, News of National Academy of Sciences of the Republic of Kazakhstan, physical-mathematical series, \textbf{227}:1 (2003), 26--31. 



%\bibitem{Kul} Kulpeshov B.Sh. On self-definable subsets of $ \aleph_0 $-categorical weakly o-minimal structures // \textit{Siberian Electronic Mathematical Reports}, Vol. 9 (2012). P. 433-438.



\end{thebibliography}

	\renewcommand{\refname}{References}
	\renewcommand{\proofname}{Proof.}
	\renewcommand{\figurename}{Fig.}
	
	
	
	
	\thispagestyle{empty}
	
	\title[Binary weakly ordered minimal theories and 1-conservative extensions]
	{\large Binary weakly ordered minimal theories and 1-conservative extensions}
	
	
	\author[Zh.T. Adil]{{\bf Zh. Adil}\protect\orcidicon{0009-0006-1377-1906}}%
	
	
	\address{Zhanar Torebekkyzy Adil
		\newline\hphantom{iii} Institute of Mathematics and Mathematical Modeling,
		\newline\hphantom{iii} 125 Pushkin St.
		\newline\hphantom{iii} 050010, Almaty, Kazakhstan}%
	\email{\textcolor{blue}{z.adil@math.kz}}%
	
	
	\author[B.S. Baizhanov]{{\bf B.S. Baizhanov}\protect\orcidicon{0000-0002-3743-7404}}%
	
	
	\address{Bektur Sembiuly Baizhanov
		\newline\hphantom{iii} Suleyman Demirel University,
		\newline\hphantom{iii} 1/1 Abylaikhan St.
		\newline\hphantom{iii} 040900, Kaskelen, Kazakhstan}%
	\email{\textcolor{blue}{baizhanov@math.kz}}%
	%
	
	\thanks{\sc Zh. Adil, B.S. Baizhanov, 
		Binary weakly ordered minimal theories and 1-conservative extensions}
	\thanks{\copyright \ 2023 Zh. Adil, B.S. Baizhanov}
	\thanks{\rm The first author is financially supported by the Science Committee of the Ministry of Science and Higher Education of
		the Republic of Kazakhstan (Grant No. AP09058169)}
	\thanks{\it Received January, 1, 2023, Published December, 31, 2023}%
	
\sloppy	
	\semrtop \vspace{1cm}
	\maketitle {\small
		\begin{quote}
			\noindent{\bf Abstract. }
			The present paper concerns application of approach introduced by David Marker and Charles I. Steinhorn for o-minimal theories \cite{MS} into binary weakly o-minimal theories. We study non-definable types using criterion of non-definability for weakly o-minimal theories \cite{BDef}. It is proved that in 1-conservative, non-conservative pair of models of binary weakly o-minimal theory there exists an irrational 1-type over small model which is not weakly orthogonal to some tuple from big model. As corollary of this result, we give an analogy of Marker-Steihnorn theorem for quite or almost o-minimal theories.
			\medskip
			
			\noindent{\bf Keywords:}  definable types, binary theory, weakly o-minimal theory, orthogonality of types, conservative extension.
		\end{quote}
	}

\bigskip


A linearly ordered structure $ \mathcal{M} = \langle M; =, <, \ldots \rangle $ is called \textit{weakly o-minimal} if any definable (with parameters) subset of $\mathcal{M}$ is a union of finitely many convex sets in $\mathcal{M}$.
A subset $A$ of a linearly ordered structure $\mathcal{M}$ is called \textit{convex} if for any $a,b\in A$ and $c\in M$ whenever $a<c<b$ we have $c\in A$. We say that theory $ T $ is weakly o-minimal if all its models are weakly o-minimal. Theory $ T $ is called \textit{binary} if every formula $ \phi(x_1, \ldots, x_n) $ is equivalent to Boolean combination of formulas in at most two free variables. We call such formulas \textit{binary}.

Suppose that $ \mathcal{M}=\langle M,\Sigma\rangle $ and $ \mathcal{N}=\langle N,\Sigma\rangle $ are models of a complete theory $ T $ such that $\mathcal M $ is an elementary submodel of $\mathcal N$ ($\mathcal M \prec \mathcal N $). For any set $ B\subset N $ we fix following notations:
\begin{center}
	$ B^+=\{z\in N\ |\ \forall b\in B, z>b\} $,
	
	$ B^-=\{z\in N\ |\ \forall b\in B, z<b\} $.	
\end{center}
Obviously, for a definable set $ B $ sets $ B^+ $ and $ B^- $ are also definable.
The following are easily derived from notations:
\begin{center}
	$ (B^+)^-=\{w\in N\ |\ \forall b\in B^+, w<b\} $,
	
	$ (B^-)^+=\{w\in N\ |\ \forall b\in B^-, w>b\} $.	
\end{center}
Notice that these notations apply for definable sets of formulas as well.

Let $ \bar y $ be a tuple $ \langle y_1, \ldots, y_n \rangle $. Denote the length of tuple $ \bar y $ by $ l(\bar y) $ ($ l(\bar y)=n $). 

Let $A\subseteq M$, $n<\omega$, $p\in S_n(A)$. The type $p$ is called \textit{definable} if for every $\phi(\bar{x}_n, \bar v)\in L(x_n)$ there exists a formula $R_{\phi}(\bar v)\in L(A)$ ($A$-formula) such that for any $\bar a\in A$ holds $\phi (\bar{x}_n, \bar a)\in p$ if and only if $\mathcal M\models R_{\phi}(\bar a)$. We say \cite{BCons} that $\mathcal N$ is \textit{$n$-conservative extension} of $\mathcal M$ ($\mathcal M \prec_{n,c} \mathcal N $), if for any $\bar \alpha \in N\setminus M$ ($l(\bar \alpha) =n$) its $n$-type is definable. $\mathcal N$ is \textit{conservative extension} of $\mathcal M$ ($\mathcal M \prec_{c} \mathcal N $), if it is $n$-conservative extension of $\mathcal M$ for any $n<\omega$.

Van den Dries \cite{Dries} proved that in an o-minimal structure $ \langle \mathbb{R}, <, \ldots\rangle $ any type $ p\in S(\mathbb{R}) $ is definable.
David Marker and Charles I. Steinhorn \cite{MS} generalized this result and proved that in an o-minimal theory if a structure $ \mathcal{M} $ is 1-conservative in $ \mathcal{N} $, then $ \mathcal{M} $ is $n$-conservative in $ \mathcal{N} $.
This theorem does not apply for weakly o-minimal theories. B.S.Baizhanov constructed \cite{BCons} an example of weakly o-minimal theory where $ \mathcal{N} $ is 1-conservative extension of  $ \mathcal{M} $ but $ \mathcal{N} $ is not 2-conservative extension of  $ \mathcal{M} $.


Let $ p$ be a type from $ S_1(A) $ with $ A\subset M $. Denote 
\vskip0.1cm $ L(p):=\{G(x) | G(x) $ is an A-definable formula such that $ G(M)<p(M)\} $ and 
\vskip0.1cm $ R(p):=\{D(x) | D(x) $ is an A-definable formula such that $ p(M)<D(M)\} $.

\begin{definition} \cite{BDef}
	Let $ \theta(\bar{z}) $ and $ H(y,\bar{z}) $ be an A-definable formulas with  $ X:=\theta(M)\cap A $. We say that the {\bf condition of left convergence} of a formula $ H(y,\bar{z}) $ on a set $ X $ or of $ \theta(\bar{z}) $ to the type $ p $ is satisfied and denote by
	\begin{center}
		$ LC\big(H(y,\bar{z}),X,p\big) $ or $ LC\big(H(y,\bar{z}),\theta(\bar{z}), p\big) $
	\end{center}
	if for any $ G(x)\in L(p) $ there exists $ \bar{a}\in X $ such that
	\vskip0.1cm $ \mathcal{M}\models \exists x (G(M)<x<H(M,\bar a)^+) $, $ H(M,\bar a)<p(M)$.
	
	We say that the {\bf condition of right convergence} of a formula $ H(y,\bar{z}) $ on a set $ X $ or of $ \theta(\bar{z}) $ to the type $ p $ is satisfied and denote by
	\begin{center}
		$ RC\big(H(y,\bar{z}),X,p\big) $ or $ RC\big(H(y,\bar{z}),\theta(\bar{z}),p\big) $
	\end{center}
	if for any $ D(x)\in R(p) $ there exists $ \bar{a}\in X $ such that
	\vskip0.1cm $ \mathcal{M}\models \exists x (H(M,\bar a)<x<D(M)) $, $ p(M)<H(M,\bar a)^+$.
	
	We also say that the {\bf condition of two-side convergence} of a formula $ H(y,\bar{z}) $ on a set $ X $ or of $ \theta(\bar{z}) $ to the type $ p $ is satisfied and denote by
	\begin{center}
		$ C\big(H(y,\bar{z}),X,p\big) $ or $ C\big(H(y,\bar{z}),\theta(\bar{z}),p\big) $
	\end{center} 
	if $ LC(H,X,p) $ and $ RC(H,X,p) $ hold simultaneously.
\end{definition}

Let $p\in S_1(A)$ where $A\subseteq M\models T$. We say \cite{BClass} that the type $p$ is \textit{right quasirational} if there exists convex $A$-definable formula $U_p(x)\in p$ such that for every sufficiently saturated model $\mathcal{N}\succ\mathcal{M}$ we have $U_p(N)^+ = p(N)^+$. We say that the type $p$ is \textit{left quasirational} if there exists convex $A$-definable formula $U_p(x)\in p$ such that for every sufficiently saturated model $\mathcal{N}\succ\mathcal{M}$ we have $U_p(N)^- = p(N)^- $. If a type is either right or left quasirational, it is called \textit{quasirational}. If a type is both right and left quasirational, it is called isolated. If a type is neither quasirational nor isolated, we say that it is \textit{irrational}.

\begin{proposition} \cite{BClass}
	Let $p$ be a 1-type over a model of a weakly o-minimal theory. The type $p$ is definable if and only if $p$ is quasirational. The type $p$ is non-definable if and only if $p$ is irrational.
\end{proposition}

\begin{theorem} \label{Th1} \cite{BDef}
	Let $ A\subset M $ and $ p\in S_1(A) $ be an irrational type. Then the following are equivalent:
	
	(i) $ p $ is non-definable;
	
	(ii) exists $ A $-definable formula $ H(y,\bar{z}) $ such that for any $ A $-definable formula $ \theta(\bar{z}) $ holds:
	\begin{center}
		$ C\big(H(y,\bar{z}),\theta(\bar{z}),p\big) \bigvee C\big(H(y,\bar{z}),\urcorner\theta(\bar{z}),p\big) $.
	\end{center}
\end{theorem}

We say that the formula $\phi(\bar x, \bar b)$ where $\bar b\in N$ \textit{divides} $C\subset N^l$ ($l$ is the length of tuple $\bar x$) if $\phi(N^l,\bar b)\cap C\neq\emptyset$ and $\urcorner\phi(N^l,\bar b)\cap C\neq\emptyset$.


\begin{definition} \cite{Shelah}
	Let $ A\subset M $. We say that type $ p\in S(A) $ is weakly orthogonal to type $ q\in S(A) $ ($ p \perp^w q $) if $ p(\bar x) \cup q(\bar y) $ define complete $ (l(\bar x) + \l(\bar y)) $-type.	
\end{definition}


Type $ p $ is \textit{not weakly orthogonal} to type $ q $ ($ p\not \perp^w q $) if there exists $ A $-definable formula $ \phi(\bar x, \bar y) $, $ \alpha \in p(M) $ and $ \beta_1 $, $ \beta_2 \in q(M) $ such that $ \beta_1\in \phi(N,\alpha) $ and $ \beta_2\not\in \phi(N,\alpha) $.
Equivalently, we say that type $ p \in S(A) $ is \textit{not weakly orthogonal} to type $ q \in S(A) $ if there exist $ \bar \alpha \in p(M) $ and $ A $-definable formula $ \phi (\bar x, \bar y) $ such that $ \phi (\bar x, \bar \alpha) $ divides $ q(M) $. 

At this point, we should recall \cite{BDef} that in weakly o-minimal theories non weakly orthogonality preserves the property of definability of 1-types, that is when $ p\not \perp^w q $, the type $p$ is definable if and only if the type $q$ is definable.



\begin{theorem} \label{Th1}
	Let $ T $ be a binary weakly o-minimal theory and $\mathcal{M}, \mathcal{N}$ be models of $T$ where $\mathcal{M}=\langle M, \Sigma\rangle$ and $\mathcal{N}=\langle N, \Sigma\rangle$ such that $ \mathcal{N} $ is a 1-conservative extension of $ \mathcal{M} $ and let $ \bar c, d\in N\setminus M $. Suppose that $ tp(\bar c/M) $ and $ tp(d/M) $ are definable and $ tp(d\cup\bar c/M) $ is non-definable. Then there exists an irrational type $r\in S_1(M)$ such that $ tp(d\cup\bar c/ M) \not \perp^w r$.
\end{theorem}

\begin{proof}
Since the type $ tp(d/M) $ is definable, it is quasirational.
Let $ A = M\cup\bar c $. Non-definability of $ tp(d\cup\bar c/M) $ implies that $ tp(d/M\cup\bar c) $ is also non-definable. By the Criterion of non-definability (Theorem \ref{Th1}), there exists $ A $-definable formula $ H(x,\bar{y},\bar{c}) $ such that for anу $ A $-definable formula $ \theta(\bar{y}, \bar{c}) $ the following holds:
\begin{center}
	$ C\big(H(x,\bar{y},\bar{c}), \theta(\bar{y}, \bar{c}),tp(d/M\cup\bar c)\big) \bigvee C\big(H(x,\bar{y},\bar{c}),\urcorner \theta(\bar{y}, \bar{c}),tp(d/M\cup\bar c)\big) $.
\end{center}


Assume that the condition of convergence holds for $ \theta(\bar{y}, \bar{c}) $ (throughout the proof we write simply "condition of convergence" \ instead of "condition of two-side convergence"). We apply method of decomposition of sets into cells by D.Macpherson, D. Marker and Ch. Steinhorn \cite{MMS}. We can partition $ \theta(M, \bar{c}) $ into a finite union of cells: $ \theta(M, \bar{c}) = \theta^1(M, \bar{c})\dot\cup \ldots \dot\cup \theta^t(M, \bar{c}) $. Since $ \theta(M, \bar{c}) $ is $ A $-definable, each cell is definable using only parameters from $ A $. The condition of convergence holds for one of these cells, say for $ \theta^s $ or its negetion. Assume that it holds for $ \theta^s $.
For simplicity purposes, we can omit index $ s $ and work on a cell $ \theta(M, \bar{c}) $.

Denote $ X:=\theta(M, \bar{c})\cap (M\cup \bar c)^{l(\bar y)} $ and let $ X = \{\bar{b}_i | i\in I\} $. It follows from the proof \cite{BDef} of the Criterion of non-definability (Theorem \ref{Th1}) that the formula $ H $ satisfies the following property: $ (H(M,\bar b,\bar c)^+)^- = H(M,\bar b,\bar c) $. We introduce order into the set $ X $ in the following sense: for any $ \bar{b}_1, \bar{b}_2 \in X $ we denote $ \bar{b}_1 <_1 \bar{b}_2 $ if and only if $ H(N,\bar b_1,\bar c)^+ \subset H(N,\bar b_2,\bar c)^+ $.

\sloppy

By the binarity of theory, every formula is equivalent to a boolean combination, and consequently, disjunctive normal form (DNF) of binary formulas. Without loss of generality, we can assume this DNF as disjoint disjunctions of conjunctions of binary formulas and their negations: 

\vskip0.1cm $ H(x,\bar y,\bar c) \equiv H_1 (x,\bar y,\bar c) \dot{\vee}\ldots\dot{\vee} H_i (x,\bar y,\bar c) \dot{\vee}\ldots \dot{\vee} H_n (x,\bar y,\bar c)$. 


\begin{lemma} \label{lemma}
	If two-side convergence holds on a formula $ H(x,\bar y,\bar c) $ on a binary theory, then it holds on at least one of the formulas from its disjunctive normal form.
\end{lemma}

The contrary of lemma \ref{lemma} is impossible. Assume that the condition of convergence does not hold on any of the formulas from DNF. Consider realisation of given formula: 
\vskip0.1cm $ H(M,\bar b,\bar c) \equiv H_1 (M,\bar b,\bar c) \dot{\cup}\ldots\dot{\cup} H_i (M,\bar b,\bar c) \dot{\cup}\ldots \dot{\cup} H_n (M,\bar b,\bar c)$. 

Since the condition of convergence does not hold on any of the formulas from given union, it does not hold on $ H(M,\bar b,\bar c) $ which contradicts the clause of lemma.

We now return to the proof of the theorem. Assume that the condition of convergence holds on $ H_i (x,\bar y,\bar c) $. Since $ H_i (x,\bar y,\bar c) $ composes a cell, by \cite{MS}, \cite{MMS}, it is monotonic in each coordinate. Assume that $ H_i (x,\bar y,\bar c) $ monotone increases in coordinate $\bar y_i$. For simplicity we can omit the index of $H_i$, then take a deeper look into this formula:

$ H (x,y_1, \ldots, y_n, c_1, \ldots, c_m) \equiv W_1 (x,y_1)\wedge\ldots \wedge W_n (x,y_n) \wedge R_1(x,c_1)\wedge\ldots\wedge R_m(x,c_m)\wedge \bigwedge\limits_{i,j\leq n} L_{i,j}(y_i,y_j) \wedge \bigwedge\limits_{k\leq n} \bigwedge\limits_{l\leq m} G_{k,l}(y_k,c_l) $.

The definable sets corresponding to the formulas $ \bigwedge\limits_{i,j\leq n} L_{i,j}(y_i,y_j) $, $ \bigwedge\limits_{k\leq n} \bigwedge\limits_{l\leq m} G_{k,l}(y_k,c_l) $ and $ R_1(x,c_1), \ldots $, $ R_m(x,c_m) $ are fixed, i.e. boundary of the formula $H$ does not depend on the behavior of these formulas. Therefore, $ H (M,\bar b, \bar c)^+ \equiv  W_i (M,b_i)^+ $ for some $ i\in \{1,\ldots, n\} $. So we look into right boundaries of the formulas $ W_1 (x,y_1), \ldots$, $ W_n (x,y_n) $. The boundary for each $ W_t $ can be expressed by the formula:

\vskip0.1cm $ B_t(x,y_t,\bar c):= W_t(x,y_t)\wedge R_1(x,c_1)\wedge\ldots\wedge R_m(x,c_m) \wedge \bigwedge\limits_{k\leq n} \bigwedge\limits_{l\leq m} G_{k,l}(y_k,c_l) $
for $ t\in \{1,\ldots, n\}$. 

Again, the condition of convergence holds on at least one of these formulas. Denote it by $B_j$. Then $ C(B_j, X_j, tp(d/M\cup\bar c)) $. Then we have 

\vskip0.1cm $X_j := \exists y_1\ldots \exists y_{j-1}\exists y_{j+1}\ldots \exists y_n \theta (y_1, \ldots y_{j-1}, N, y_{j+1}, \ldots, y_n, \bar c) \cap (M\cup \bar c)$.  The set $ X_j $ can be partitioned into finite union of cells: $ X_j = X_j^1 \dot\cup \ldots \dot\cup X_j^s $. The condition of convergence holds at least on one of them, say $X_j^l$: $ C(B_j, X_j^l, tp(d/M\cup\bar c)) $. Since a cell of dimension 1 defines a convex set in a weakly o-minimal theory, $X_j^l$ is a convex set where $ W_j $ monotone increases. Then $ X_j^l $ contains an irrational type $r\in S_1(M)$. Since the type of $\bar c$ is definable, $ tp(\bar c/M) \perp^w r $. 

Let $ r'= tp(\delta/M\cup\bar c) $ where $ \delta\in r(N) $. Notice that $ r'(N) = r(N) $. Then $ \forall b\in X_j^l $, $ \models W_j(d,b,\bar c) \Rightarrow b>r'(N) $ and $ \models \urcorner W_j(d,b,\bar c) \Rightarrow b<r'(N) $. Thus, $ tp(d/M\cup\bar c)\not \perp^w r' $ which implies that $ tp(d\cup\bar c/M)\not \perp^w r $.


\end{proof}


Let $\mathcal N$ be a sufficiently saturated model of the theory $T$, $A\subset N$, $p, q\in S_1(A)$. We say \cite{BOtypes} that the type $p$ is \textit{not almost orthogonal} to $q$ ($ p\not \perp^a q $) if there exists $A$-definable formula $\phi(x,y)$ such that $\forall \alpha\in p(N)$, $\exists \beta_1, \beta_2\in q(N)$ so that $\beta_1<\phi(N,\alpha)<\beta_2$, $ \emptyset \neq \phi(N,\alpha)\subset q(N) $ holds. The set $A\subset N\models T$ of weakly o-minimal theory $T$ is said to be \textit{almost o-minimal} if for every $p,q\in S_1(A)$ we have $q\not \perp^w p \Longleftrightarrow q \not \perp^a p$. We say that weakly o-minimal theory is \textit{almost o-minimal} if any of its sets is almost o-minimal.

We say \cite{Kul1} that the type $p$ is \textit{not quite orthogonal} to the type $q$ ($ p\not \perp^q q $) if there exists $A$-definable bijection $f: p(M)\rightarrow q(M)$. We say that weakly o-minimal theory is \textit{quite o-minimal} if the notions of weakly and quite orthogonality of 1-types coincide.


\begin{theorem}
	Let $T$ be a binary weakly ordered minimal theory and $\mathcal M \prec_{1,c} \mathcal N $.  Suppose $\mathcal M \nprec_c \mathcal N $ and, consequently, there exists $\bar \alpha \in N\setminus M$ such that $tp(\bar \alpha|M)$ is non-definable, length of tuple $\bar \alpha$ is minimal. Then there exists irrational $1$-type over $M$, $r\in S_1(M)$ such that
	$tp(\bar \alpha|M)\not \perp^w r$,   $tp(\bar \alpha|M) \perp^a r$.
\end{theorem}

\begin{corollary}[Analogy of Marker-Steinhorn Theorem]
	Let $T$ be a binary weakly ordered minimal theory. If $T$ is quite o-minimal or almost o-minimal, then for any elementary pair of models $\mathcal M \prec \mathcal N $ the following holds:
	$\mathcal M \prec_{1,c} \mathcal N $ implies $\mathcal M \prec_{c} \mathcal N $.
\end{corollary}

\begin{proof}
	Consider the irrational type $r\in S_1(M)$ and $\delta \in r(N)$ from Theorem \ref{Th1}. Again let $r'=tp(\delta/ M\cup\bar c)$. Then $tp(d/M\cup\bar c)\not \perp^w r' $. For quite and almost o-minimal theories the notions of not quite and not almost orthogonality coincide with not weakly orthogonality. So there exists $A$-definable formula $\varphi(x,d,\bar c)$ such that $\varphi(N,d,\bar c)\subset r(N)$, that is $\mathcal{N}\models \exists x\varphi(x,d,\bar c)$ which implies that $\delta\in N$. Since the type $ tp(\delta/ M) $ is irrational, it is non-definable. It contradicts the assumption that $\mathcal M \prec_{1,c} \mathcal N $, so there does not exist a non-definable type over $M$ realised in $N$.
\end{proof}






\bigskip

\end{document}